Author's response to reviews

Title: The prognostic relevance of interactions between venous invasion, lymph node involvement and distant metastases in renal cell carcinoma after radical nephrectomy

Authors:

Dragomir P Zubac (zubac.dragomir2423@gmail.com)
Leif Bostad (leif.bostad@helse-bergen.no)
Tomas Seidal (tomas.seidal@lthalland.se)
Tore Wentzel-Larsen (tore.wentzel-larsen@helse-bergen.no)
Svein A Haukaas (svein.haukaas@helse-bergen.no)

Version: 4 Date: 6 December 2008

Author's response to reviews:

We have made the formatting changes requested. The e-mail addresses have been changed as requested. 'Legend to figure' has been changed to 'Figure legend'. The heading 'Tables' has been added to the tables section. The figure has been cropped according to the guidelines. We have carefully read the manuscript and corrected a few typographical errors.